Author's response to reviews

Title: Effects of Work-Related Factors on the Breastfeeding Behavior of Working Mothers in a Taiwanese Semiconductor Manufacturer: A Cross-Sectional Survey

Authors:

Yi Chun Chen (yichun@tmu.edu.tw)
Ya-Chi Wu (ycwu@cde.org.tw)
Wei-Chu Chie (weichu@ntu.edu.tw)

Version: 5 Date: 13 June 2006

Author's response to reviews: see over
BMC-series journals
Middlesex House
34-42 Cleveland Street
London W1T 4LB

Re: 5893251118278401 'Effects of work related factors on breastfeeding behavior of working mothers in Taiwan's Semiconductor Manufacturer: A cross sectional survey'

June 13, 2006

Dear Editor,

Thank you very much for your e-mail on June 11, 2006 of the comments for our manuscript 5893251118278401. We have revised our manuscript according to your comments:

1. All the blue highlighting was removed.

2. Methods: The description of consent was clarified. We obtained oral consent. Written consent forms were not required. (page 5)

3. Authors’ contributions: We have changed authors’ names to initials and added the declaration of final approval.

4. Questionnaire: we will upload it separately as an additional file.

Thank you again for your kind attention and we are looking forward to hearing from you soon.

Sincerely yours,

Wei-Chu Chie, M.D., Ph.D.
Professor, School of Public Health and Graduate Institute of Preventive Medicine
College of Public Health, National Taiwan University
Room 520, 17 Xuzhou Road
Taipei 10055
Taiwan

Tel: 886-2-23920460
Fax: 886-2-23920456
E-mail: weichu@ntu.edu.tw